EV/18-July-05

Week 30  starting 27 June 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  25 July 05 @: 7/25/2005 3:08:55 PM
	11
	A1
	MBAR3371
	Prep 0 Scan Bench
	299.2

	8
	A2
	MBAL2170
	ICS 4 Final connection
	298.49

	1
	B1
	SSS173
	PT 6.2 Training 2
	1.90

	7
	B2
	SSS195
	Prep 0 Scan Bench
	299.11

	4
	C1
	MBBR3360
	PT 12 Warm Up
	278.49

	5
	C2
	MBBL3121
	Prep 5 Cool Down
	298.31

	9
	D1
	SSS152
	PT 13.2 Resist. Meas.
	299.02

	12
	D2
	FREE
	Bench MAINTENANCE
	298.76

	2
	E1
	MBBL3115
	PT 6.1 Training 1
	5.37

	6
	E2
	MBAL1207
	Prep 1 Lyre Test
	298.49

	10
	F1
	MBBR3336
	PT 13.2 Resist. Meas.
	298.58

	3
	F2
	MBAR3367
	Prep 5 Cool Down
	88.98


	


   Temperature data are taken from TBSM18 site

 
Coordinator this week: Marco;MTM support 16-3602

13 complete Magnet Tests cycles
9 dipoles( MM )+ 4 SSS( 1 with special Thermal cycle without MM & SSW )
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· General Points :

· Special S4 (Florence)

· Special Tests D2 : Silver shoes repaired, Q4 S4.

· Rotary Switch cluster B difficult to used.

· Rotary Switch for Corrector powering.

· AOB

From 7/17/2005 to 7/23/2005

	-
	Bench
	Magnet
	Shafts
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for

	1
	TBA1
	MBBL2189(2) 
	[image: image1.png]



	12-07-2005 
	30 h.
	25 h.
	50 h.
	17 h.
	0 h.
	17-07-2005 
	122 h.
	Thermal cycle

	2
	TBA1
	MBBL2189(2) 
	[image: image2.png]



	17-07-2005 
	3 h.
	27 h.
	22 h.
	13 h.
	8 h.
	20-07-2005 
	73 h.
	Stripping

	3
	TBB1
	SSS199 
	[image: image3.png]



	13-07-2005 
	29 h.
	133 h.
	29 h.
	10 h.
	16 h.
	22-07-2005 
	217 h.
	Stripping

	4
	TBB2
	SSS204 
	[image: image4.png]



	08-07-2005 
	172 h.
	13 h.
	34 h.
	12 h.
	11 h.
	17-07-2005 
	242 h.
	Stand By

	5
	TBC1
	MBAR3369 
	[image: image5.png]



	16-07-2005 
	34 h.
	50 h.
	39 h.
	6 h.
	13 h.
	22-07-2005 
	142 h.
	Stripping

	6
	TBC2
	MBAL1246 
	[image: image6.png]



	19-07-2005 
	14 h.
	48 h.
	17 h.
	15 h.
	20 h.
	23-07-2005 
	114 h.
	Stripping

	7
	TBC2
	MBBL3085 
	[image: image7.png]



	15-07-2005 
	26 h.
	24 h.
	14 h.
	13 h.
	8 h.
	18-07-2005 
	85 h.
	Stripping

	8
	TBD1
	SSS217 
	[image: image8.png]



	12-07-2005 
	33 h.
	25 h.
	135 h.
	3 h.
	16 h.
	21-07-2005 
	212 h.
	Stand By

	9
	TBD2
	SSS617 
	[image: image9.png]



	15-07-2005 
	52 h.
	16 h.
	4 h.
	15 h.
	9 h.
	19-07-2005 
	96 h.
	Thermal cycle

	10
	TBE1
	MBAL3109 
	[image: image10.png]



	16-07-2005 
	45 h.
	54 h.
	17 h.
	15 h.
	9 h.
	22-07-2005 
	140 h.
	Stripping

	11
	TBE2
	MBAR3373 
	[image: image11.png]



	13-07-2005 
	34 h.
	25 h.
	27 h.
	21 h.
	9 h.
	18-07-2005 
	116 h.
	Stripping

	12
	TBF1
	MBBL2206 
	[image: image12.png]



	13-07-2005 
	35 h.
	30 h.
	15 h.
	26 h.
	8 h.
	18-07-2005 
	114 h.
	Stripping

	13
	TBF2
	MBBL3126 
	[image: image13.png]



	16-07-2005 
	44 h.
	46 h.
	15 h.
	15 h.
	10 h.
	22-07-2005 
	130 h.
	Stripping


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	TC Q1
	TC Q2

	1
	MBBL2189
	9160.2 A.
(D1-L)
	11076 A.
(D2-L)
	11904 A.
(D2-L)
	12024.1 A.
(D2_U)
	11098.9 A.
(No Data)
	12426 A.
(D1 - U)
	12655 A.
(D2 - U)
	12307 A.
(D2 - U)
	12667 A.
(D2_U)
	12820 A.
(D2-U)
	12700 A.
(D2-U)
	[image: image14.png]




	2
	MBBL2189
	12016.8 A.
(D1-U)
	12828 A.
(D2-U)
	[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	
	

	3
	SSS199
	11733 A.
(QI-34)
	12401 A.
(QE-12)
	12756 A.
(QE_12)
	[image: image16.png]



	-
	-
	-
	-
	-
	-
	
	

	4
	SSS204
	11450 A.
(QI-34)
	11607 A.
(QE_34)
	12337 A.
(QE-34)
	12850 A.
[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	
	

	5
	MBAR3369
	10255 A.
(D1-U)
	12393.5 A.
(D1_U)
	12648 A.
(D2-U)
	[image: image18.png]



	-
	-
	-
	-
	-
	-
	
	

	6
	MBAL1246
	11810 A.
(D1-L)
	12019 A.
(D2-L)
	12230 A.
(D1-L)
	[image: image19.png]



	-
	-
	-
	-
	-
	-
	
	

	7
	MBBL3085
	12600.9 A.
(D1-U)
	12850 A.
[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	
	

	8
	SSS217
	11813.1 A.
(QE-34)
	12615.5 A.
(QI_12)
	12836 A.
(QI-12)
	12850 A.
[image: image21.jpg]



	12850 A.
[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	

	9
	SSS617
	5890 A.
[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	10
	MBAL3109
	11687.1 A.
(D2_L)
	12850 A.
[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	
	

	11
	MBAR3373
	10700 A.
(D1-U)
	11531 A.
(D2-U)
	12852.2 A.
(D1-U)
	12850 A.
[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	
	

	12
	MBBL2206
	11926 A.
(D2-L)
	12773 A.
(D1-L)
	12850 A.
[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-
	
	

	13
	MBBL3126
	11843 A.
(D1-U)
	12246 A.
(D2-U)
	12594.2 A.
(D2-L)
	[image: image27.png]



	-
	-
	-
	-
	-
	-
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _19_M

Tuesday _19_N

Tuesday _19_N


	ALL

Bench

TBE1

TBC1
	Cryogenic Supervision lost.

PLCSE communication error.

Intercon lights do not flicker when we switch over to X2. Unable to run test PT4 due to interlocks, QheaterC_x1, PowerQH_OK. Potaim card red light on V_supra_D and Vsum_Ext_s.
	ACR

AB/CO


	
	ACR piquet has been called by cryo team, and asked to wait 8:00. Antonio Tovar is not in call.

Close Tema, reset server crate, ok after 
Call Equip Support for Intercon box problem

	Wednesday _20_M

Wednesday _20_M

Wednesday _20_N


	TBB1

TBD1

TBA2

TBB1
	HV could not be launched because of wrong cryo phase interlock, 

further power tests has been stopped because of no signal from LQA.

Powering corrector GH-IJ are not possible to perform, because of a different cabling.
	MTM

MTM
	
	Reinstallation of the programs was not done properly on this cluster, now it is solved. Richard
Equipment Support has been informed.

Richard has been informed during morning, and till the new rotary switch is not install, equipment support will switch between B1 to B2.

	Thusday _21_N


	TBF1
	Trigger on Potaim card Vbus_Ext & Vbus_Int. Unable to launch PT1 (partIII) AC compensation test on TBF1. Richard informed
	MTM
	
	Problem is caused by the INTERCON box not switching some of the main voltage taps of the dipole. I have shorted one cable on INTERCON so that we can go on with the tests on TBF1. But we can not switch it to X2, it needs an intervention to put INTERCON in normal configuration. Richard.



	Friday_22_M

Friday_22_M


	TBC2

TBF
	PT4, quench during SPA, below 70 amperes. 
Problem identified, it's due to 2 random different reason: 1st, the potaim card Number 2 was not working properly (it has benn changed). 2nd, It seems that we have a problem with the Power converter behaviour (AP/PO informed, they should perform some test on monday). SPA during ramping for first training
Richard has interchanged the chassis for minimum energy with the one of cluster D which is not used on this cluster. 
	MTM

MTM
	
	The "variac" is destroyed and needs to be replaced. Richard.

For the moment operators must perform an dico reset before each power test.Test aborted. 



	Friday _22_A


	TBF
	Switch between F1 to F2 or reverse way
	MTM
	
	Richard has put some sticker to help operation team to change this cabling during week end.



	Friday _22_N


	TBB2
	Files not transferred to the 'Storage/vol12' folder for SSS036_5. Data could not be validated for PT5.1 & PT6.1


	AB/CO
	
	?

	Saturday_23_M


	TBA1
	AC Signal wire above CFB partly broken. Needs to be soldered.
	MTM
	
	?

	Saturday_23_M


	TBE1
	ICS informed and show that the wire for the V clamp connection is broken, not possible to connect it


	MTM
	
	?

	Saturday_23_A


	TBB1
	IAP clamp connection of correctors could not be launched because of self generated trigger. ICS open sleeve to check the connection.


	ICS
	
	ICS found wrong connection and done the right connection. IAP clamp connection of correctors has been done successfully.

	Saturday_23_A


	TBF1
	MM mobile rack XMT.MY71 was not working. Amplifier error got during AutoTest. The Rack has been changed but there was problem with Scanning. Both the MOBIC are not working properly. Equipment support has been informed.


	MTM
	
	?

	Saturday_23_N


	TBD1
	SSS152 HV at cold failed on (M(X)-D M(X)-F correctors and Sextupoles)
	?
	
	Magnet will be signed stand by.

	Sunday_24_M


	TBF1
	during MM3 (Full loadline) ramping : SPA.


	
	
	Test was relaunched.

	Sunday _24_N


	TBD1
	Communication failure between sunmta35 and PLCSE. The popup "Contacting PLCSE " comes up every 1-2 min. The same problem occured on 23/07/2005 also on TBD1.

Then HV could not be started because of "wrong Cryo phase"


	AB/CO
	
	It was solved by switching OFF the PLCSE and then restarting.

	Monday _25_N


	TBF2
	Problem with HV test on bench F2 magnet 3367. at 3050v step of "cold initial before test connected to CFB" Dipole quench heaters versus ground current reading was about 10mA and error was generated. connections checked and everything seemed OK. 
	MTM
	
	so rack changed from MY95 to MY93 and test relaunched. No errors appeared. Rack MY95 marked with sign signalling fault and that it should not be used until checked by eq supp.


